Author's response to reviews

Title: Solid variant of aneurysmal bone cyst of the heel; a case report

Authors:

   Joanna Lekka (Jlekka@yahoo.gr)
   Theofani Gavressea (fangav@261.gr)
   Gabriella Stanc-Giannakopoulos (gabidr1999@yahoo.com)
   Nikolaos Demertzis (NDemertzis@yahoo.gr)

Version: 10 Date: 30 November 2010

Author's response to reviews: see over
Author's response to reviews

Solid variant of aneurysmal bone cyst of the heel: a case report

J. Lekka (Jlekka@yahoo.gr)
Th. Gavresea (fangav@yahoo.gr)
G.Stanc-Giannakopoulos (gabidr1999@yahoo.com)
N. Demertzis (Ndemerzis@yahoo.gr)

Author response to review: see over

Restructured abstract:
Editor's report

1) In the Abstract and Case Presentation section, please replace the nationality of the patient with his ethnicity.

Changes have been made.

Version 9: 30/11/2010